RESEARCH Open Access



Perspectives of underweight people with eating disorders on receiving Imagery Rescripting trauma treatment: a qualitative study of their experiences

Marieke C. ten Napel-Schutz^{1,2*}, Simona Karbouniaris^{3,4}, Suzanne H. W. Mares¹, Arnoud Arntz⁵ and Tineke A. Abma⁶

Abstract

Background: The prognosis for underweight individuals with an eating disorder (ED) and posttraumatic stress disorder (PTSD) is worse than for their peers without these comorbid symptoms. This qualitative study explores the experiences of trauma-focused Imagery Rescripting (ImRs) therapy of underweight inpatients being treated for an ED.

Objective: To test the feasibility and to improve ImRs by understanding the experiences and perspectives of people with an ED and PTSD who, when underweight, received ImRs as an adjunct to their inpatient ED treatment.

Method: To explore how underweight people with an ED experience and perceive ImRs, we used a qualitative study design involving semi-structured interviews with 12 participants. After analysis, the data were summarized and classified within a thematic framework that focused on experiences and improving the ImRs method.

Results: The thematic analysis resulted in the following 6 main themes; (1) Expectations of ImRs; (2) Ability to participate in ImRs; (3) Effect of ImRs; (4) Experience of ImRs technique; (5) Conditions under which ImRs is given; (6) In depth-analysis. The results show that despite the fear of disappointment the participants appreciate addressing the PTSD and ED symptoms simultaneously. Further, results showed that it had been possible for them to attend ImRs but that the effects of ImRs were not uniformly perceived. Also, participants indicated that a caring context is important and that ImRs should not be scheduled immediately before a meal. Finally, the treatment generated hope.

Conclusions: The findings of this study demonstrated the feasibility of the integration of ImRs trauma treatment for individuals who are being treated in an ED inpatient treatment setting, and are in contrast to standard practice where the focus of inpatient treatment has been ED-symptom improvement without comprehensively addressing past traumatic experiences during an underweight phase.

Trial registration International Clinical Trials Registry Platform (ICTRP) (NTR6094). Date of registration 09/23/2016. https://trialsearch.who.int/Trial2.aspx?TrialID=NTR6094

Keywords: Eating disorders, Underweight, Anorexia nervosa, PTSD, Trauma treatment, ImRs, Qualitative research

¹ Department of Eating Disorders (Amarum), GGNet Mental Health, Warnsveld, The Netherlands Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third partial in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

^{*}Correspondence: m.tennapel@ggnet.nl

Plain English summary

A proportion of those with eating disorders have also experienced traumatic events and ongoing symptoms of PTSD such as re-experiencing of the trauma and nightmares. We implemented an innovative trauma intervention called Imagery Rescripting (ImRs) to explore whether for those undergoing inpatient treatment for an eating disorder (in an underweight phase), it would be possible to treat the various trauma-related symptoms as well as the eating problems. Since this has not been investigated before, we asked the participants in this study to recount their experiences. Twelve participants who were underweight, reported a past history of trauma and were in an inpatient eating disorder treatment program participated in ImRs therapy intervention. One of these participant did not engage in the ImRs therapy because she discontinued the inpatient ED treatment. Analysis of interviews with these participants found that -although they were reluctant before the start of the treatment- the ImRs treatment during their inpatient admission had given them hope again. They added that it was important to have support from group members, sociotherapists and therapists. They shared a number of ways that the ImRs treatment could be adapted to people with eating disorders. Their experiences indicated that given these factors it was possible to treat PTSD during an underweight phase. This is important: until now, treatment for eating disorders has not specifically been trauma-focused and these tips have scope to improve the ImRs intervention and eating disorder treatment more broadly in the future.

Background

Post-traumatic stress disorder (PTSD) symptomatology is a specific risk factor for the development of an eating disorder (ED) such as Anorexia Nervosa (AN) [1]. Researchers have hypothesized that certain ED behaviours may help a person to reduce, avoid, or escape from PTSD symptoms [1, 2, 3]. Between 10 and 47% [4, 5] of people with AN are diagnosed with PTSD—a group, particularly the patients treated in higher levels of care, who are known not only to be harder to treat than those without PTSD [3, 6, 7], but also to have a greater number of other co-morbid symptoms (such as anxiety and depression symptoms, obsessive—compulsive symptoms, and interpersonal symptoms), and also lower self-esteem and poorer quality of life [2, 6, 8, 9, 10].

It is therefore unsurprising that several studies have shown that this group of people respond worse to treatments, quit treatment prematurely, or are more likely to relapse, than people who have AN without PTSD [2, 3]. Patient organizations and several researchers have therefore long proposed that treatment outcome might be improved by treating trauma at an earlier stage, i.e. during an underweight phase (IxtaNoa, personal communication, May 2012) [2, 3]). Some studies have been done on treating trauma symptoms in addition to ED symptoms [9, 11, 12]. But very little research has investigated the options for treating trauma symptoms during an underweight phase.

In the clinical tradition to date, psychotherapeutic trauma-focused treatment is not initiated in people who are underweight [13, 14]. This is because it is assumed that, during periods of underweight, cognitive functioning is suppressed [15, 16, 17], too few emotions are experienced [18], and emotion regulation is poor [19, 20]. But although poor functioning in these areas is assumed to

obstruct effective trauma-focused treatment, there is no research consensus on poor cognitive functioning, failure to experience emotions, and the associations between body mass index (BMI) and emotion-regulation skills.

To seek a solution to the lack of evidence and the need for appropriate and helpful treatment, a broader study by ten Napel-Schutz et al. [21] incorporated a quantitative multiple baseline case series study involving ten participants with PTSD, an ED, and a BMI of between 14 and 16.5. The results of the latter showed that people with PTSD could be treated effectively when underweight. The following symptoms were all reduced: PTSD symptoms, emotional experiences of disgust, shame, guilt and anger; negative cognitions about the body, the self and the world; and difficulties with emotion regulation. The eating pathology also became less severe [21].

The treatment method used in this case series study was the evidence-based treatment ImRs for PTSD (Additional file 1: box S1) [22, 23, 24, 25]. Although ImRs had not yet been used with people who were underweight, it was used here because earlier studies had shown (1) that another trauma method (EMDR) requires sufficient physical and mental stability because it can be an invasive procedure [26], (2) that ImRs refrain from full exposure to traumatic memories and therefore may present fewer difficulties than treatments that do rely on full exposure [27], (3) that participants in the study of [28] found ImRs to be less invasive than exposure, and (4) that here was less dropout with ImR than with pure Imagery Exposure and this was seen as an indication of the less invasive experience of ImRs [29], and (5) that it achieved good results with perceived anger, guilt, and shame [28].

Because ImRs had not previously been offered for PTSD during an underweight state and during clinical admission, it is helpful to know what kind of experiences participants have, how they perceive the treatment, and what, from their perspective, can be done to improve it in ways that will better attune it to their needs. To gain insight into the experiences of trauma treatment with ImRs in people being treated in an inpatient ED setting, the present qualitative study therefore aimed to explore individuals': (1) Expectations of the treatment; (2) experiences of IMRs treatment, including any perceived shifts in cognitive functioning, emotional experiences and their regulation; and (3) perspectives on the feasibility of treating trauma when they were also underweight.

Method

Setting

Parallel to the inpatient treatment they were receiving for their ED while underweight, participants in the present study received twelve ImRs sessions. All the participants had been diagnosed with PTSD and were female; their demographic and clinical data are listed in Additional file 1: Table S1. The interviews were held two weeks after the end of ImRs treatment, at that time, 1 participant had already stopped inpatient ED treatment while the other participants were still receiving it.

Ethics

Approval for the intervention study, and thus also for our qualitative study, was obtained from the medical ethical committee at the University of Amsterdam (2016-CP-7111). Informed consent was signed by all participants. Pseudonymity was guaranteed, and was achieved by deleting the participants' names and personal characteristics from the data, and by not sharing the transcripts with the therapists and other peers who had inpatient treatment for their ED. All statements in this article have been disconnected from the people who made them. The verbatim transcripts and recordings will be destroyed five years after publication of this article. Although, conceivably, participants may be able to link some statements to fellow participants, all committed themselves to confidentiality agreements at the start of clinical treatment.

All participants were told that it was possible and permissible not to answer questions. The researcher/first author is a clinical psychologist who works at the same institute but had no therapeutic relationship with the participants. Before agreeing to participate, participants had received an announcement and an explanation of the study, accompanied by an invitation to participate.

Design

As we wished to improve the treatment by learning more about the participants' personal experiences and perceptions, and also by knowing what meaning they would attribute to ImRs, we chose a qualitative study with interviews. We chose a semi-structured approach that was

more descriptive than phenomenological or interpretive, as this would be better suited to our aim of eliciting information that would help to improve the treatment.

Data collection

Data collection involved semi-structured interviews with participants [30]. These were based on a pre-defined topic list, whose initial draft was first discussed with an ImRs professional, with whom it was then confirmed. The validity of the topic list was then checked in three individual interviews with participants. After the addition of several topics, the definitive topic list was then established (see Additional file 1: Table S2).

Each interview started with an introduction to the interview and with questions on sociodemographic data, after which the topics on the topic list were discussed (see Additional file 1: Table S2).

The interviews were held in a therapy room at the institute where the ED treatment took place. Each interview was led, guided and structured by the same person (the first author), who ensured that all topics were discussed. She also took fieldnotes. To make sure all topics were addressed and the answers were properly understood, she asked questions and listened actively, connected information in her fieldnotes, and, in collaboration with the participants, she tried to understand as best as possible what the participant meant [30]. The interviewer was as open as possible to the experiences of the interviewee, an intention that was emphasized during the interview. Care was taken to prevent common pitfalls, such as outside interruptions, stage fright, awkward questions, jumping from one subject to the other, and the temptation to counsel respondents [30].

All interviews were recorded on an mp3 player after the participants' approval had been received. They were elaborated verbatim. A summary was written afterwards. The interviews had no fixed duration, but ended when all topics had been addressed and no new information emerged. Their length ranged from 30 to 90 min.

Sampling

We used an iterative process in which the topic list was adjusted after the first three interviews. In qualitative research it is customary to select a relatively small group of respondents until data saturation is attained [30]. As most interviews achieved saturation on most of the predefined and some extra topics, it was possible to make valid statements on these topics. However, some topics were so person-related that saturation was not achieved. The repeated themes create the opportunity to gain a thorough understanding of experiences [31]. Twelve people were invited for the interview: the ten completers of the ImRs treatment, one person who had dropped out

of treatment (after three weeks of ImRs, and one person who did not start ImRs because she could not sustain the clinical ED treatment. She discontinued clinical ED treatment before the start of ImRs because she had difficulty with participating in the ED treatment group, she was diagnosed with an autism spectrum disorder. She was interviewed to find out if the upcoming ImRs treatment was a reason for her to quit. One completer, had hallucinations in addition to PTSD that were trauma related and had no bizarre content. These secondary psychotic symptoms were not considered contraindications for participation in the ImRs. The hallucinations did not get worse but also did not diminish after the first ImRs sessions so anti-psychotic medication was given for a short period of time without result. The patient herself was very eager to complete the ImRs so she was included in the results as a completer.

Analysis

In line with the framework analysis developed by the National Centre for Social Research, we summarized and classified the data within the following thematic framework: (1) The researcher familiarized herself with the data by listening to the recordings and re-reading the fieldnotes and transcripts; (2) A framework (coding scheme) was developed, based partly on the themes that guided the interviews, and partly on the issues that arose in the transcripts; (3) The text fragments in the transcripts were coded and labelled using the coding scheme; (4) Indexing was done intersubjectively by two researchers, who worked systematically on themes and sub-themes; (5) The associations between these themes and sub-themes were then established; and (6) Working independently, the researchers coded and categorized the themes and sub-themes they contained. If disagreements arose on coding, they were discussed until consensus was reached. With the support of the qualitative data-analysis program atlas-ti, the data was charted, i.e. rearranged by theme.

It was particularly useful that the second researcher not only had a research background but was also an expert-by-experience: previous research has shown that such a dual identity can be advantageous. When a study is conducted only by professionals, it is easier for sight to be lost of the patient's perspective [32, 33, 34].

Quality procedure

As a validity check we used respondent validation, in which each participant was sent the verbatim transcript

and a short analytical summary of their interview by secured email [30, 35, 36]. Participants were asked if the interpretations corresponded with their perspective, and if they expressed what they had meant. This respondent validation showed that all participants agreed that the summary of their interview was accurate. One participant commented that during the interview she looked back positively on the ImRs treatment, but that her opinion had changed by the time respondent validation took place. She indicated that she had not managed to (1) Indicate that conditions at the time of ImRs were not optimal, and (2) that she now felt that she had needed longer/extended ImRs treatment.

Trauma-focused ImRs

Participants were offered twelve 90-min individual ImRs trauma sessions over six-weeks [25]. ImRs is a technique that attempts to change the meaning of an experience by looking at traumatic childhood memories in a different way. This experiential therapeutic technique uses imagery and imagination to imaginatively intervene in traumatic memories. The trauma memory is activated, i.e. the patient imagines the start of the traumatic memory and when there is enough emotional activation, the rescripting starts. The therapist (first six sessions), and in the last six sessions the participant (each from their current perspective), then rescripts the traumatic experience in a more desirable direction, while imagining this new script in the liveliest way possible. For a more detailed treatment description see Additional file 1: box S1. Four participants experienced physical assault, 5 participants experienced sexual assault, 2 participants had other unwanted or uncomfortable sexual experiences, 1 participant experienced combat or exposure to a war zone in the military and 1 participant experienced another very stressful event. One participant experienced 2 onetime traumatic experiences, the other participants experienced more complex trauma related to prolonged and severe traumatization at a younger age [21].

ED treatment

In the clinic the participants followed a programme for their ED that consisted of inpatient treatment five days a week. They spent their weekends at home. The programme used cognitive behaviour-change methods and focused on weight gain and on the factors that were maintaining the ED. When they were underweight, participants committed to a weekly weight gain of 700 g. The therapeutic programme was offered entirely in group form and consisted of (1) three supervised main meals and snacks daily, (2) cognitive behavioural therapy (focused on the thoughts and behaviours associated with the ED), (3) eating-behaviour therapy (focused on the

¹ National Centre for Social Research. https://www.natcen.ac.uk/. Accessed 27 Feb 2022.

eating diary), (4) body and movement therapy, (5) psychoeducation, (6) progress and goal discussion, (7) psychodynamic group therapy (addressed the factors that caused and/or perpetuated the psychological problems and focused on both the individuals in the group and the interaction between group members), and (8) occupational therapy. Additionally, five times during their stay, participants were given opportunities to take part in a parent-and-partner group.

Results

The topic-list questions produced five fixed themes. Close reading then produced a further 25 sub-themes. Through an iterative process, an additional theme with 3 sub-themes was added. The 28 sub-themes describe a total of 81 categories.

Below, the results are described on the basis of the six themes and 27 sub-themes. Additional file 1: Table S3 provides an overview of the themes, subthemes, categories and illustrative quotes. Additional file 1: Table S4 provides an overview of tips and suggestions from a participants' perspective.

Expectations

Overall, all participants had appreciated the opportunity to participate in this study, particularly because of the opportunity to address the symptoms of the ED and the PTSD simultaneously while they were underweight. Many had gone through a long search for adequate treatment. "I was very happy there was a clinic somewhere that was willing to offer trauma treatment even though I was underweight." Although three participants had had positive expectations of ImRs, the others had been coloured by earlier negative experiences of EMDR treatment, fearing disappointment, no longer dared to expect anything and had chosen to take the trauma treatment without expectations.

As well as indicating that it was difficult, a priori, to get a good idea of ImRs, two participants had also had doubts about its effectiveness: "Because the therapist explained that you can make up all sorts of things in your imagination, I wondered something like 'is this going to work on my trauma...?."

With regard to their preparation for the study, all participants indicated that they had received sufficient information. (See Tips and Suggestions Additional file 1: Table S4, point 4.1; Illustrative Quotes, Additional file 1: Table S3, points 1.1 and 1.2.)

Ability to participate in ImRs

In the interviews, participants talked about the ability to participate in ImRs in relation to their level of concentration, their ability to experience feelings, their ability to regulate emotions, gaining weight, their ability to imagine, their physical effects, their trauma topicality and openness about their trauma.

Ability to control attention and to concentrate during ImRs

With one exception, all participants indicated that their attention span had been sufficient to do the ImRs, even though they were underweight and had other co-morbid disorders (Attention Deficit Hyperactivity Disorder, Tourette Syndrome etc.). "Yes, I was generally able to concentrate sufficiently." The exception indicated that, after a full day of therapy, she had been too tired for the ImRs treatment. Although she dropped out after three sessions of ImRs, she had experienced a positive result on her PTSD symptoms, but had been too tired the next day to follow the clinical treatment properly. "I managed, but it was such an effort that on the following day I could no longer concentrate on the therapy in the clinic."

Ten participants reported adequate concentration. Of these, six participants indicated that three factors had made it difficult to maintain their concentration throughout the ImRs session (90 min): the time of day, the length of the ImRs session, and the many distractions caused by ambient noise. "You hear things in the hallway, so it's difficult to maintain your concentration." The therapists found solutions to these distractions: a quiet room, turning off the light, shortening the sessions, and finding a more suitable time for the ImRs. Despite the difficulties, three participants reported that well-coordinated individual person-oriented support had eventually made it possible for them to concentrate properly. (See Illustrative Quotes, Additional file 1: Table S3, point 2.1; Tips and Suggestions, Additional file 1: Table S4, point 4.2.)

Ability to feel

The interviews showed clearly that almost all (10) participants experienced enough emotions to do ImRs: "Yes, I'm still a bit overwhelmed: I could feel them very well." One person said she had tried to avoid her feelings, but her therapist saw from her body signals that the tension was building up. Especially at the end of the sessions, after ImRs, she had a lot of feelings.

In the period in which ImRs was provided, two participants experienced a change in their ability to feel, even when their body weight increased. "No, I don't really feel as if it changed very much." Although one participant indicated that she had been able to access her feelings better during the second six sessions, she attributed this to the method. "From the moment I started to intervene myself it became a bit easier." (Illustrative Quotes, Additional file 1: Table S3, point 2.2; Tips and Suggestions, Additional file 1: Table S4, point 4.2.)

Ability to regulate emotions

Seven participants indicated that they found it scary to feel emotions they had not felt for an extended period or had done their best to avoid. One participant said: "I feel everything again, and that's just what I wanted to get rid of." Another: "I'm not used to feeling emotions at all any more—I've always pushed them away, and it's scary when they're there again."

During ImRs they were asked to start feeling these emotions. As this differed slightly from the ED treatment, in which they tried to let go of their old coping strategy of regulating emotions through ED behavior and thoughts, it evoked not only the fear of getting overwhelmed and lingering in the emotion, but also the experience of managing not to be flooded by emotions. "And then I was crying really hard and then it was okay too—not that I got stuck in it, because that was what I was really afraid of: getting stuck in it." They indicated that ImRs helped them more than previous methods to learn to regulate their emotions. They also reported that it helped that they could do ImRs in a safe clinical treatment environment where emotions were welcomed, and where they could cry, talk with peers and sociotherapists; or just go for a walk. (Illustrative Quotes, Additional file 1: Table S3, point 2.3; Tips and Suggestions, Additional file 1: Table S4, point 4.2.)

Ability to gain weight, and the role of weight and underweight

Participants had different experiences of being able to gain 700 g body weight per week during ImRs. While two participants said that the trauma treatment required too much energy for them to be able to continue the growth curve, all the others were able to manage both. However, as their trauma symptoms seemed to become worse or more noticeable when they gained weight, these participants were unanimous that help was needed to continue to put on weight while simultaneously handling the trauma symptoms. "Yes, [weight gain] did make me feel worse, and I really felt that you shouldn't wait to help me until I'm at a healthy weight, as that would make it very difficult for me actually to attain that weight." For this reason, participants experienced the support they received at mealtimes during the ImRs period as essential to gaining or maintaining a stable weight during the ImRs treatment. "So, yeah: seeing I've previously had trauma treatment that gave no support to eating—where I actually went on hunger strike—yeah, I'm glad [ImRs] was done in a clinical setting." (Illustrative Quotes, Additional file 1: Table S3, point 2.4; Tips and Suggestions, Additional file 1: Table S4; 4.2.)

Ability to imagine

Many individual differences emerged regarding participants' ability to work with imagination. Four could easily imagine situations and even marvelled at it. One participant noted that she was a visual thinker, so this technique suited her very well: "I was surprised how well I could imagine it." Two participants had to get used to it, after which it worked well. Two participants mentioned that, if you are an image thinker, the pictures really come into your mind, are extra vivid, and that if you think more rationally, it can be hard to imagine things that are not realistic or have not happened. For one person, imaging the therapist in the image did not work, as it was too far removed from reality. It had then helped her that she and her therapist decided to adapt the script and stay closer to reality. (Illustrative Quotes, Additional file 1: Table S3, point 2.5; Tips and Suggestions, Additional file 1: Table S4; 4.2.)

Ability to engage in ImRs: physical effects

Four participants found that ImRs caused many physical reactions, such as headaches, sweating, palpitations, fatigue, pain in the legs, paralysis of limbs, and an inability to move the legs. "I had a lot of physical reactions during and after ImRs—it was quite intense." Although one participant stopped ImRs due to fatigue, the physical reactions were not a reason for the other participants to discontinue it. Someone verbalized that the physical reactions decreased because they had learned to reassure themselves.

The participants with physical reactions indicated the importance of announcing in advance that multiple physical reactions could occur. As well as indicating the importance of the therapist remaining calm and looking for person-oriented entrances, they made several suggestions for making it possible for a participant to continue ImRs treatment despite the physical reactions; in the event of limb failure, for example, it helped to stand during the session. It also helped to have a wheelchair ready, not to schedule the ImRs sessions two days in a row, and to make it possible for participants to remain in the room after a session until they felt grounded again.

In summary, although participants experienced many physical reactions, these were a reason for hardly any of them to stop ImRs. (Illustrative Quotes, Additional file 1: Table S3, point 2.6; Tips and Suggestions, Additional file 1: Table S4, point 4.2).

Ability to engage in ImRs: trauma topicality and openness about trauma

Regarding the topicality of the trauma—in other words, the extent to which they were still troubled by it—it emerged that two participants were not yet fully in a safe environment outside the clinic (i.e. in their private lives). Three participants indicated that their traumas were no longer topical but that tensions could still be evoked, as they still had contacts with the perpetrator. For example, one perpetrator occasionally visited over the weekend; another continued to send messages; and as one group of perpetrators still lived in the same area, the participant occasionally ran into them.

Two participants indicated that, with a therapist they knew barely or not at all, they had difficulty being open about the trauma they had experienced. They sometimes found it difficult to share at all. One person indicated that she had started to write down her whole trauma story not only for the therapist to read, but also for any future therapists; this would make it unnecessary to tell her story again. Two participants indicated that they were unable to tell the whole story about the events, one reason being that they did not dare to complete certain memories in their mind. Another participant gradually increased her degree of openness, starting with the less difficult trauma images, and eventually discussing the more difficult images only in the final ImRs sessions.

Three participants indicated that the more often there was openness, the easier it went. If it was possible to be completely open, this was experienced as pleasant. But overall, all participants developed sufficient trust during the sessions to provide enough disclosure about their traumas. (Illustrative Quotes, Additional file 1: Table S3, point 2.7; Tips and Suggestions, Additional file 1: Table S4, point 4.2.)

In summary, participants indicated that, given well-coordinated person-oriented adjustments and a joint search for ways to maintain the treatment with the clinical treatment team and therapists, the following had been possible: to attend the treatment, to apply the imagining technique, to tolerate the physical effects, and to disclose their traumas sufficiently. They could do this with sufficient concentration, emotional experiences, and emotion regulation; and they were able to maintain or gain weight.

Perceived effect of ImRs

In the interviews, participants talked about their perceived effects of ImRs in terms of negative emotions, eating disorder symptoms, PTSD symptoms, self-compassion, internalization of ImRs and expected effects of ImRs in the future.

Perceived effect of ImRs on negative emotions

Although two participants found it difficult to feel anger, two participants noted that this improved during the ImRs sessions. Three participants indicated that they started to feel angrier about what had been done to them. Another participant noted that she succeeded in feeling less angry, and two participants in directing their anger less towards themselves but also towards the perpetrator or perpetrators. "First I was only angry at myself, and now I'm also angry at the people who did that to me." It emerged that six participants had been able to re-frame the trauma. (Illustrative Quotes, Additional file 1: Table S3, point 3.1.)

Different experiences emerged regarding the effect of ImRs on guilt. Four participants started to feel less guilty or were able to reframe the traumatic experience. "What happened really shouldn't have happened, and I was able to experience that it had not been my fault." Two participants noticed no difference in their feelings of guilt. However, there was also someone (who also experienced psychotic symptoms) who started to feel more guilty. "If only it had happened like that, if only I had made a different choice." (Illustrative Quotes, Additional file 1: Table S3, point 3.2).

With regard to the perceived effect of ImRs on shame, two participants reported that shame stayed out of the picture during treatment, and, even though the qualitative questionnaires had explicit questions on it, that the therapist had not asked about it directly. Typically, shame concerned questions of what the participants would have liked to do with the perpetrator or perpetrators, or shame about their own physical reactions to ImRs. "If you have so much shame, you're not going to talk about it." It was suggested that the therapist should ask explicitly about feelings of shame. "If we'd been asked, I think it would have come up more often. Otherwise, if you're so ashamed, you're not going to tell them." It also helped two participants that they could do whatever they wanted to do in their thoughts, and that they did not have to speak out loud. The therapist could support this process by taking a robust approach to it, by keeping calm, and by agreeing with participants what they should include in the imaging in order to reduce the feelings of shame. (Illustrative Quotes, Additional file 1: Table S3, point 3.7; Tips and Suggestions, Additional file 1: Table S4, point 4.3.)

Perceived effect of ImRs on ED symptoms

With regard to the effect of ImRs on eating, eight participants stated that, due to the emotions they had experienced through ImRs, it had been difficult to eat a meal after the session on the day of the ImRs. After ImRs, participants typically reported "feeling dirty", having "too much tension", having "too many sensory re-experiences",

and having "food triggers"; or "not being hungry when very emotional," and "not being able to eat because of a lump in my throat." One participant reported difficulties also the day after ImRs. Four participants also indicated that they used "not eating" on the day of ImRs to avoid or distract themselves from emotions such as sadness and anger, and that the ED helped to push the emotions away. "ImRs makes it hard to get through the day. So, to suppress those feelings a bit again, I did tend to cut down on food." (Illustrative Quotes, Additional file 1: Table S3, point 3.3).

It is noteworthy that the topic of negative body perception was not addressed very much during the ImRs sessions. As one participant linked her negative body awareness to the ED and not to the trauma, she had no hope of change through ImRs. However, another participant who did discuss her negative body awareness during ImRs also experienced no effect of this on her body awareness: "My body still feels dirty and all that." Two participants indicated the importance of explicitly discussing any negative body perceptions that may have originated from the traumatic event. (Illustrative Quotes, Additional file 1: Table S3, point 3.4; Tips and Suggestions, Additional file 1: Table S4, point 4.3).

Perceived effect of ImRs on PTSD symptoms

Participants reported variously that ImRs had had positive, neutral, and negative experiences on their PTSD symptoms.

With regard to its positive effect on PTSD symptoms, someone expressed surprise at how easily the brain could be changed. In her case, when she thought back to the unpleasant event, she now saw the therapist in the picture. Three participants stated that they had managed to stay in the here and now better by thinking about the rescripted situation. "Now I can also say to myself: it's not now, it was then. And then I try to see a bit of the rescripted scene in front of me. So it will end well." Three participants also reported that their reactions to re-experiencing symptoms became less intense after a few ImRs sessions, and that their physical reactions, such as palpitations and sweating, and their nightmares, flashbacks, and emotional reactions to ImRs, had decreased. "No more crying all the time."

One participant indicated that she had noticed little effect on her PTSD symptoms. She did not feel that she was thinking about her trauma less, or was less sensitive to stimuli, and still felt very vulnerable, unstable, and emotional. "No, I don't think anything has changed regarding the degree of those symptoms."

Two participants reported a temporary aggravation of their PTSD symptoms, consisting of an increase in re-experiences, some of which became more intense during the period or on the day of the ImRs. During a

re-experience they did not manage to go back to what had been done in the rescripting. They indicated that when a re-experience began, they consciously tried to call up the image from the session. They could then see this new image for a moment, after which the old image took over. " Sometimes, when images start coming up, I consciously try to call up an image from those sessions. But often I see it just for a moment and then it's gone again—the other image, the real one, flies over and covers it." Three participants had a re-experience or dissociation during the ImRs sessions. "During ImRs I often found myself going into a kind of express train, so to speak, and the film started running for me, so I was actually in a re-experience I couldn't escape from." After the sessions, two participants suffered a lot from nightmares." And certainly, in the nights after the sessions I also had a lot of nightmares." (Illustrative Quotes, Additional file 1: Table S3, point 3.5).

Five participants indicated that ImRs initiated a grieving process whereby, through rescripting, they realized what protection or care they had missed at the time of the traumatic event. "[It's a] process of mourning... and you think...'if only I'd had that." (Illustrative Quotes, Additional file 1: Table S3, point 3.6).

Perceived effect of ImRs on self-compassion, internalization and its expected future effect

The treatment was also found to influence the development of self-compassion. Self-compassion seemed to have increased in three participants, though a few (2) did not experience this. For example, while one participant did not at first know what to say to reassure herself, her ability to do so improved during the session. In contrast, two other participants indicated that they had not learned to look at themselves with more compassion. (Illustrative Quotes, Additional file 1: Table S3, point 3.10).

At the time of the interviews, it seemed that seven participants had not yet (fully) internalized the ImRs method, and were unsure whether they could expect any effect from it over time. However, two participants were already able to use it independently. (Illustrative Quotes Additional file 1: Table S3, point 3.5).

When asked if they expected any future effect from the ImRs treatment, participants had various ideas. Five participants indicated that they did not know: "I've no idea what to expect." Two participants thought it would have not any further effect, while four participants thought the effect they had experienced would develop further: "I hope so, and definitely think it will do something." They added that while it was the start of a nice change, it was not enough yet: "I don't know if this will get rid of it

completely, but I do think it will do something." (Illustrative Quotes, Additional file 1: Table S3, point 3.11).

Experiences with the ImRs technique

In the interviews, participants discussed their experience with the ImRs technique imagination with rescripting and their experiences with this technique in the first and second six ImRs sessions.

Imagination and rescripting

Seven participants reported that the imagining had gone well and were amazed how well they had been able to experience it. "Surprised how well I could imagine it and how it felt exactly like the time of the trauma." Five of the participants also reported that they liked the fact that they could keep control. "I found it quite tough, but also a very nice treatment, as you get control and keep it." One participant said that when she realized that she herself could determine the intensity of the tension she felt, she started to dread it less. "I was not looking forward to it, and after the first session I was dreading it even more. And after that it got less, because at a certain point I realized that I could determine how great I allowed the tension to become—to what extent I went back into that memory. I liked that."

On the other hand, two participants indicated that the method made it possible to avoid unpleasant feelings or images. For example, one participant indicated that there were unfinished images to which she did not dare to go. During the rescripting, she did not have to finish these images. (Illustrative Quotes, Additional file 1: Table S3; 4.1, Tips and Suggestions, Additional file 1: Table S4; 4.1.)

The first six sessions

In the first six sessions, the therapist steps into the image and intervenes in coordination with the patient. Two participants reported that this was difficult in the beginning, as they had never shared certain events before. Seven participants indicated that it was nice that the therapist had intervened during the first six sessions, that they had not had to do it alone, and that there had been an example: "When X did it, I was pleased that she acted a bit more angrily than I could." But at the same time, two people also found it confronting that no-one had helped in the past.

Five participants felt that rescripting with the use of fantasy felt unrealistic and rather crazy, which also could limit the imaging. They were aware during rescripting that the rewritten situation did not correspond with reality. One person became aware, especially after the rescripting, that if a rewritten and imagined situation was a bit bizarre, they sometimes experienced it as

unrealistic. However, another participant found it impossible to let the therapist enter the picture because it was too unrealistic for her, which is why she imagined the help from her parents.

During rescripting, participants were asked what their needs were. Two participants experienced the question of need to be difficult or confronting, but two participants experienced the questions also as a question that was very nice. "I found it quite difficult, as I very often didn't know what my need was. But I liked the fact that she had asked, because the question helped me realise much more strongly that I could think of something I actually wanted." However, two participants also stated that identifying the need became easier along the way. "As the sessions progressed, I was increasingly able to communicate what I needed." (Illustrative Quotes, Additional file 1: Table S3, point 4.2; Tips and Suggestions, Additional file 1: Table S4, point 4.4.)

The last six sessions

In the last six sessions, the participant rescripts herself. Here the experience was mixed. Five participants found these sessions to be nicer than the ones guided by the therapist: they felt more empowered, were more in control, and could create their own narrative. "I liked the second part better because I could do it myself. And after that I felt more powerful." On the other hand, three participants found it more difficult to take the lead. Sometimes, for example, the connection with the child in the image was completely broken. In two other cases, they did not dare to speak to the perpetrator out loud, failed in their attempts to comfort the child, or found it unpleasant to have to do it alone. "Having to step into the picture myself, as an adult—that took a lot of effort, especially because I noticed that I did not see my little self as myself. As if I were helping another child." At the end of each ImRs sessions, the therapist asked participants to bring their attention back to the present. Two participants reported that this was not an easy process, and that they still felt "floaty" for a while. (Illustrative Quotes, Additional file 1: Table S3, point 4.3; Tips and Suggestions, Additional file 1: Table S4, point 4.4.)

Contextual conditions

The ImRs was offered at the time of inpatient admission for the ED treatment. In the interviews, the participants shared their experiences with the support in the clinical setting, with treating multiple participants at the same time, with beliefs about ImRs in a clinical setting, with the support during meals, about the combination of ImRs in a clinical setting and about the number, the timing and the scheduling of the ImRs sessions.

Clinical setting; support

Nine participants particularly appreciated ImRs treatment in the context of the ED facilities. They gave several reasons for this, such as the distraction, the structure, the relaxation, and the fact that it felt too heavy to return home after ImRs. The support after ImRs was particularly appreciated, as it had made it possible to talk to the sociotherapists, group members, or roommates; or to go for a walk, to avoid getting stuck in unpleasant feelings, or to prevent withdrawing. "And they [the sociotherapists] also helped me with reexperiences or dissociations, so to speak-they helped me through it. And after that they made sure I was completely here again and safe." However, one person indicated that she needed more support, as she had mostly coped alone. She had found it difficult to ask for support. There were also two participants who just wanted to be alone for a while after ImRs and had no specific need for support. (Illustrative Quotes, Additional file 1: Table S3, point 5.1; Tips and Suggestions, Additional file 1: Table S4, point 4.5.)

Clinical setting; several participants at the same time

It happened twice during the study that two participants in the same ED treatment group received ImRs in the same period. One participant reported that a fellow patient's re-experiences had triggered her own re-experiences. Two participants indicated that, at times, they felt guilty when they needed the sociotherapists' support at the same time as someone else, as this prevented the rest of the clinical group from getting attention (Illustrative Quotes, Additional file 1: Table S3, point 5.1.2; Tips and Suggestions, Additional file 1: Table S4, point 4.5).

Clinical setting; belief about ImRs

Two participants indicated that at times it seemed that participants and clinic staff had the idea or belief that undergoing ImRs treatment was very intense. A participant who did *not* think it was so intense therefore became uncertain about whether she was entitled to ImRs because she did not experience it like that (Illustrative Quotes, Additional file 1: Table S3, point 5.1.3; Tips and Suggestions, Additional file 1: Table S4, point 4.5).

Clinical setting; mealtime support

By offering ImRs in the context of the ED facility, there was active support around meals. Almost everyone (9) found this to be supportive. "If I hadn't been in this environment, I think I would just have stopped eating." (Illustrative Quotes, Additional file 1: Table S3, point 5.1.4; Tips and Suggestions, Additional file 1: Table S4, point 4.5).

Clinical setting in combination with ImRs

Two participants found it difficult to undergo a six-week programme that combined two weekly 90-min ImRs sessions with the inpatient ED-treatment programme. "Found it quite tough to have one of those sessions twice a week in the inpatient programme—that really took a lot of energy out of me, so yes, I did find it tough." Five participants who had managed this full programme were positive about the fact there had been sufficient closure time after the rescripting. The participant who stopped after three ImRs sessions indicated that it had been too hard for her to attend ImRs as well as the clinical ED programme: "Unfortunately that was just a bit too much-I did benefit from it, but I was so tired that I no longer had the energy to continue." (Illustrative Quotes, Additional file 1: Table S3, point 5.2.1; Tips and Suggestions, Additional file 1: Table S4, point 4.5.)

Number of ImRs sessions

Opinions about the number of ImRs sessions differed widely. While some participants thought that 12 sessions had not been enough, others thought 12 was enough or even too many. Three participants who had experienced multiple traumas, however, indicated that 12 sessions may have been too few: "I had quite a lot of memories I wanted to discuss, [and] didn't get to do them all. I needed more than 12 sessions." (Illustrative Quotes, Additional file 1: Table S3, point 5.2.2; Tips and Suggestions, Additional file 1: Table S4, point 4.5.)

Timing of ImRs

A six-week naturalistic baseline had been followed by a random baseline period. Six participants felt that the start of ImRs had been well timed, because they needed some time to get used to the ED treatment. "Just a few weeks to get used to the programme in the group". Two participants also indicated that if their weight had been lower than the weight specified in the study's inclusion criteria, ImRs might not have been successful. "If my weight had been lower, ImRs probably wouldn't have been possible. I was in a phase of not feeling. If I'd had to start feeling then, I don't know if I could have, and I wouldn't have dared to make the step towards." One participant indicated that, for her, the onset of the ImRs treatment should have been earlier. She had taken a pause from the clinical programme two weeks before starting ImRs because the PTSD symptoms became too severe. "I didn't like it because the trauma symptoms had become so severe, and nothing was being done about them yet. The only thing being stressed [i.e., during the ED treatment] was eating. While eating had become extremely difficult because of all the re-experiences." (Illustrative Quotes, Additional file 1: Table S3, point 5.2.3; Tips and Suggestions, Additional file 1: Table S4, point 4.5.)

Scheduling of ImRs

In principle, the ImRs sessions were scheduled for 3:15 pm, i.e. after the clinical therapy blocks. However, for scheduling reasons, it was sometimes necessary for a participant, and sometimes for the therapist, to move the session to the morning. While seven participants indicated that the time of the session did not greatly matter, they preferred it not to be planned immediately before a meal. "A meal after the session wasn't doable—it invariably ended up in the bucket." It had been intended to spread the two ImRs sessions over the week, but circumstances did not always allow this. On the one hand this was experienced as pleasant for one participant, as she was then "done with it." But on the other hand, she experienced it as very heavy. (Illustrative Quotes, Additional file 1: Table S3, point 5.2.4; Tips and Suggestions, Additional file 1: Table S4, point 4.5.)

Themes emerged through the semi-structured interview

During the interviews, participants identified three other themes 1) hope/ perspective, 2) effect of ImRs on awareness, and 3) compassionate approach.

Hope/perspective

While five participants expressed they were feeling hopeless before the start of their trauma treatment (Additional file 1: Table S3; 1.1.3), they saw hope in the fact that research was being done into the possibility of treating trauma in underweight people. Three participants who had hoped that treatment might bring change were pleasantly surprised when their expectations were exceeded. "Yes, I thought it might suit me, but I didn't think it would be as good as it was."

After the ImRs sessions, the five participants who had not dared to hope for a positive effect before the study seemed to have become more hopeful. They dared to look to the future: "Yes, I'm very glad I did this. I had been at a loss, and thought, "yes, this is the only chance I have to get the combination [of ED and PTSD];" and "I'm very glad it went this way, because [...] if I'd done ImRs in a non-clinical setting, it would have gone really wrong." The participant who experienced no positive effects from the ImRs indicated that she was nonetheless glad she had been able to complete the 12 sessions. There had been no recurrence of the sense of failure that had followed her previous trauma treatment, when too many re-experiences had forced her to stop.

Afterwards, the three participants specified that it was helpful that the therapists had continued to hold out hope during this new treatment modality. In summary, it appears that everyone had been given hope by their experience of ImRs when underweight—some because they had succeeded in completing treatment, others because they noticed small changes, and others due to significant decreases in PTSD symptoms. (Illustrative Quotes, Additional file 1: Table S3, point 6.1; Tips and Suggestions, Additional file 1: Table S4, point 4.6).

Effect of ImRs on awareness

The interviews indicated that, in various areas, ImRs had initiated a process of awareness that had not previously been present. This had enabled six participants to reframe the trauma: "I was able to give a different colouring to what had taken place."

There are various examples of this awareness. One participant realized that she had PTSD. Others recognized the way things were connected, the severity of their trauma, the level of their tension, the lack of certain support figures, or their suppression of unpleasant memories. "That whole piece of becoming aware of something that I've tucked away for so long under a block of concrete and landmines. It's... I compare it to a pimple that has burst. And now it's burst, and the pus can come out." In three participants, the awareness also triggered a grieving process. For two other participants, the awareness brought the hope of future improvement and insight. (Illustrative Quotes, Additional file 1: Table S3, point 6.2).

Compassionate approach

All the interviews testified to the importance the participants attached to a compassionate approach—something in which therapeutic attitude, group setting and ImRs method played a role.

One participant indicated that she had not had a good connect with her therapist, and thus changed therapists after two sessions. All other participants indicated that their therapeutic contact had been very compassionate. Comments included "Very nice," "very pleasant," "I felt taken seriously," "she had expertise," and "I felt she really listened to me."

One participant indicated that therapists needed to realize that few such participants are securely attached individuals who have had only a single trauma. This makes it even more important for therapists to have a compassionate attitude. "I'm just not used to it [getting help]: I always told myself I didn't deserve it. Then suddenly it's very strange that someone does help [during ImRs]." Nine participants felt that the clinical group setting was also supportive, as compassionate support and understanding were both available. "The group was very helpful, and really supportive and wished me well when I went to ImRs."

The ImRs method also contributed to not feeling alone. The therapist promoted this by stepping into the picture to help, which is experienced as both pleasant and compassionate. Because the therapist was constantly asking what someone needs, nine participants felt truly seen and understood, and saw the extent and reliability with which the therapist sympathized and helped. As a result, they saw this method as much less protocol-based than other, more standardized therapies such as EMDR. (Illustrative Quotes, Additional file 1: Table S3, point 6.3; Tips and Suggestions, Additional file 1: Table S4, point 4.5).

Discussion

The clinical tradition to date has held that psychotherapeutic trauma-focused treatment should not be initiated in people who are underweight [13, 14]. The underlying assumption is that the conditions necessary for effective treatment cannot be met in such people, as cognitive functioning may be suppressed [15, 16, 17], not enough emotions are experienced [18] and emotion regulation is poor [19, 20]. The ability to experience emotions is important in trauma treatments. Olofsson et al. [37] showed that exposure to emotional and sensory trauma provided a good outcome.

However, the experience of participants in this study did not align with this presumption. They reported sufficient cognitive functioning, and also experienced a range of emotions. Despite their initial fears, they were also able to regulate these emotions, even though it was very scary to experience them again after such long periods of suppression. This emphasizes the importance of receiving caring support from group members, sociotherapists, and therapists. Other factors are equally important: to maintain or gain weight, to apply the imagining technique, to tolerate physical reactions, to be sufficiently open about the traumas they have experienced, a joint search for ways to remain in treatment and if necessary, to organize person-oriented adjustments by the therapist.

ImRs requires patients to use their imagination. Research showed that alexithymia (the inability to identify or describe one's own feelings, externally oriented thinking) is common in ED and PTSD [38, 39, 40]. This difficulty to put words to feelings (or fantasize) may complicate the use of imagination. Future research should focus on whether there are limitations or modifications needed for this specific group with alexithymia to be treated with ImRs. In the interviews, three participants indicated two weeks after the ImRs sessions that they had not experienced any improvement with regard to the trauma symptoms. Neither, we note, did these participants experience a significant worsening of their PTSD and eating disorder symptoms. It is essential that the information given to future participants includes a warning about this range of possible responses.

Participants provided several tips for the use of ImRs in individuals with an ED during an underweight phase. One of the many tips participants provided is that unlike with other psychopathology groups, ImRs should not be planned immediately before a meal, and that the holding of the clinical setting is important with regard to maintaining the eating pattern. Inpatient admission helped with ImRs because there was extensive support immediately after the sessions to prevent participants from relapsing into disordered eating and weight loss after processing the trauma in the intervention. In line with Brewerton et al. [8] and Reyes-Rodriques et al. [3], it emerges that integrated treatment of ED and trauma is important. For follow-up research, it is important to take this aspect from extensive support during inpatient admission into account when considering trauma treatment for uED patients.

Straus et al. [41] defined compassion on the basis of five elements: "recognizing suffering, understanding the universality of human suffering, feeling for the person suffering, tolerating uncomfortable feelings, and motivation to act/acting to alleviate suffering". Participants experienced ImRs as a highly suitable way of achieving these qualities, as the therapist is present in imagination at the traumatic event and can empathize with the situation very well. This is consistent with the findings of Bosch [42] and Olofsson [43] on the importance of the positive connection (with insecurely attached patients) with the therapist and getting encouragement for patients. Participants who had experienced EMDR indicated that ImRs is better suited to a compassionate approach. This may also explain why ImRs was completed by ten of the eleven participants who started it, the only dropout being one who experienced such positive effects (besides considerable fatigue) that she first wanted to work on her ED and then to continue the ImRs. For follow-up research, we recommend that a well-controlled comparative study is conducted between these two methods.

It is noteworthy that most participants specified that a process has been initiated which included a process of awareness—for some a grieving process—in which a process of increased self-compassion was combined with the feeling of not being left alone. These processes are well-known phases, and, for some participants, necessary steps on the road to recovery. Their great importance became apparent when it emerged that this group had felt very hopeless before the start of treatment. All had had a history of ineffective treatments followed by a long search for proper treatment; few had dared to have any expectations of ImRs. They reported that ImRs had given them hope again—thereby increasing their chance of better treatment outcomes [44, 45, 46].

Our study showed that the therapists did not always address feelings of shame and negative body perception—that if a therapist did not ask about them explicitly, these issues were not discussed. However it is important to address the body [37]. To ensure that they are treated, and that therapists are alert to the participants' and their own avoidance [47], it is important that any feelings of shame and negative body perception arising from the trauma are discussed explicitly. Conceivably, they were not discussed in our study because they were related to experiences that are not regarded as trauma in the DSM definition of PTSD. It might therefore be helpful if ImRs covered all early experiences associated with these feelings and not solely the trauma defined by the DSM. Possibly therapists do not discuss these issues because of uncertainties about how to integrate trauma treatment with an ED treatment, because of concerns about worsening ED symptoms, concerns that it would interfere with ED treatment or worsen self-harm [12]. Follow-up research should focus on the effects of ImRs in treating all early experiences associated with them and give attention to addressing existing concerns.

With regard to other effects, we found that the participants did not all experience decrease of anger, guilt, and PTSD symptoms two weeks after the end of ImRs treatment. This is in line with the multiple baseline study, which found a significant reduction in these variables only after three months [21]. As it is important to realize in this population that significant reductions in symptoms occur only after a few months, we recommend that participants in further research are interviewed after three months. While this interval may be necessary because participants need a little more time to work on the ED symptoms at the same time, it may also be that their underweight state is a contributing factor. And although it may be possible to start a certain process when underweight, a decrease in PTSD symptoms can be experienced only once the weight has normalized further. Follow-up research should establish whether ED symptoms (including BMI) are related to a decrease in PTSD symptoms after ImRs treatment.

Implications

The results of our study imply that according to patients it is possible to treat PTSD with ImRs during an underweight phase in a clinical ED setting with adequate caring support, person-oriented adjustments and not schedule the trauma treatment immediately before a meal. The most important implication is that there are opportunities for this until now very difficult to treat group of patients.

The results of our study provides grounds for followup research such as a multi baseline case series study with more ImRs sessions or a well-controlled comparative study between EMDR and ImRs integrated with an ED treatment, in a caring and supportive setting. Several variants for an RCT are also recommended like an RCT of PTSD treatment in parallel compared with usual treatment of just the ED, or parallel vs sequential treatment of the ED and the PTSD. It is important to add a costeffectiveness study, to take into account the experienced compassion in relation to drop out numbers and effectiveness of the treatment, and to include the diagnosis of alexithymia. Furthermore, we recommended follow-up measurements after three months and 1 year and establish whether ED symptoms (including BMI) are related to a decrease in PTSD symptoms after trauma treatment. For follow-up research in outpatient ED settings, it is essential to organize extensive support for care after the trauma treatment sessions and support at meals.

Strengths and limitations of this study

Although the researcher (i.e. the first author) is well versed in the treatments and can interpret the experiences participants have during therapy, there is a risk that her professional view may have led her to attribute meanings to the participants' experiences. The participation of the second researcher was intended to rectify this by providing experiential expertise and safeguarding the participants' perspectives in the analysis and the interpretations were checked with the participants who thought they were accurate. Another limitation is that the interviews were conducted two weeks after the end of the ImRs, and thus provided no insight into the participants' opinions and experiences in the longer term. Neither was the timing of the interview linked to the decrease in ED symptoms. Finally, since this was a single case series study, the number of interviews was limited.

Conclusion

Our study aimed to improve the trauma treatment and to test the feasibility of trauma treatment provided to people with an ED who, while still underweight, underwent ImRs during clinical treatment for their ED. A second aim was to understand these participants' experiences of ImRs and their perspectives on it.

Our results show that, thanks to the opportunity to have simultaneous treatment of their ED and PTSD in the safe environment of the clinic, participants dared to have trauma treatment despite the feelings of hopelessness that had remained after their experiences of earlier treatment. An important component of daring to feel and daring to continue eating despite the fear of the emotions they might feel was the support and compassion of peers and the clinical team. Even though they had been underweight, participants indicated that they

had been able to undergo treatment because their ability to concentrate, and to experience and regulate emotions, had all been sufficient.

The study also demonstrates the importance of properly establishing in advance and during the ImRs treatment what each participant will need to complete trauma treatment. Such preparation and adjustment contributed to their experience of ImRs as a pleasant, compassionate method, in which their sense of being on their own had been reduced by the therapist's empathy and personal focus. Two weeks after the ImRs treatment, participants also indicated that they had gained more hope about processing their trauma and ED symptoms. With regard to the long run, however, they were uncertain whether they could expect more effect from the ImRs than from other treatments. It is essential that the information given to future participants includes an explanation of possible delayed effects.

To better tailor treatment to participants' needs, ImRs should not be scheduled directly before a meal. However, shame and negative body experiences should be elicited explicitly, and everything an individual needs to sustain treatment should be carefully adjusted and monitored.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40337-022-00712-9.

Additional file 1. Table S1. Interviewees' demographic and clinical data.

Acknowledgements

We would like to thank the participants for their valuable contribution to this study, and to acknowledge GGNet, Amarum for its provision of financial and material support.

Author contributions

MtN initiated, organized, and directed the entire investigation. She also analysed and interpreted the participant data and wrote the manuscript. SK analysed and interpreted the participant data and co-wrote the manuscript. SM co-wrote the manuscript. AA made a substantial contribution to the design of the research, and substantively revised the manuscript. TA made a substantial contribution to designing the research, interpreting the participant data and co-writing the manuscript. All authors read and approved all substantially modified versions of the manuscript, and approved the version that was submitted. All authors have agreed both to be personally accountable for their own contributions and have ensured that questions related to the accuracy or integrity of any part of the work, even those in which they were not personally involved, have been appropriately investigated, resolved, and documented in the literature. The author read and approved the final manuscript.

Funding

The author declares that no funding.

Availability of data and materials

Reasonable requests for data will be considered by the authors under the condition that the European General Data Protection Regulation (GSPR) is guaranteed for these sensitive participant data, and an appropriate analytic plan is included.

Declarations

Ethics approval and consent to participate

Ethical approval was obtained from the Ethics Review Board of the University of Amsterdam (reference number 2016-CP-7111).

Consent for publication

Consent for publication is available from all participants.

Competing interests

A. Arntz occasionally provides training in ImRs. The financial remuneration he receives for this is paid to the University of Amsterdam to support research. The authors have no conflict of interest to declare.

Author details

¹Department of Eating Disorders (Amarum), GGNet Mental Health, Warnsveld, The Netherlands. ²Radboud Centre Social Sciences, Radboud University, Nijmegen, The Netherlands. ³Utrecht University of Applied Sciences, Utrecht, The Netherlands. ⁴Leiden University Medical Centre, Leiden, The Netherlands. ⁵Department of Clinical Psychology, University of Amsterdam, Amsterdam, The Netherlands. ⁶Department of Public Health and Primary Care, Leiden University Medical Centre, Leyden Academy on Vitality and Ageing, Leiden, The Netherlands.

Received: 21 July 2022 Accepted: 17 November 2022 Published online: 30 November 2022

References

- Briere J, Scott C. Assessment of trauma symptoms in eating-disordered populations. Eat Disord. 2007;15(4):347–58. https://doi.org/10.1080/ 10640260701454360.
- Brewerton TD. Eating disorders, trauma, and comorbidity: focus on PTSD. Eat Disord. 2007;15(4):285–304. https://doi.org/10.1080/1064026070 1454311.
- Reyes-Rodríguez ML, Von Holle A, Ulman TF, Thornton LM, Klump KL, Brandt H, Crawford S, Fichter MM, Halmi KA, Huber T, Johnson C, Jones I, Kaplan AS, Mitchell JE, Strober M, Treasure J, Woodside DB, Berrettini WH, Kaye WH, Bulik CM. Posttraumatic stress disorder in anorexia nervosa. Psychosom Med. 2011;73:6. https://doi.org/10.1097/PSY.0b013e31822232bb.
- Gleaves DH, Eberenz KP, May MC. Scope and significance of posttraumatic symptomatology among women hospitalized for an eating disorder. Int J Eat Disord. 1998;24(2):147–56. https://doi.org/10.1002/(SICI) 1098-108X(199809)24:2%3c147::AID-EAT4%3e3.0.CO:2-E.
- Tagay S, Schlegl S, Senf W. Traumatic events, posttraumatic stress symptomatology and somatoform symptoms in eating disorder patients. Eur Eat Disord Rev. 2010;18:2. https://doi.org/10.1002/erv.972.
- Carter JC, Bewell C, Blackmore E, Woodside DB. The impact of childhood sexual abuse in anorexia nervosa. Child Abuse Negl. 2006;2006(30):3. https://doi.org/10.1016/j.chiabu.2005.09.004.
- Rodríguez M, Pérez V, Garciá Y. Impact of traumatic experiences and violent acts upon response to treatment of a sample of Colombian women with eating disorders. Int J Eat Disord. 2005;37:4. https://doi.org/10.1002/ eat.20091.
- Brewerton TD, Perlman MM, Gavidia I, Suro G, Genet J, Bunnell DW. The association of traumatic events and posttraumatic stress disorder with greater eating disorder and comorbid symptom severity in residential eating disorder treatment centers. Int J Eat Disord. 2020;53(12):2061–6. https://doi.org/10.1002/eat.23401.
- Mitchell KS, Mazzeo SE, Schlesinger MR, Brewerton TD, Smith BN. Comorbidity of partial and subthreshold PTSD among men and women with eating disorders in the national comorbidity survey-replication study. Int J Rat Disord. 2012;45:3. https://doi.org/10.1002/eat.20965.
- Scharff A, Ortiz SN, Forrest LN, Smith AR. Comparing the clinical presentation of eating disorder patients with and without trauma history and/or comorbid PTSD. Eat Disord. 2021;29(1):88–102. https://doi.org/10.1080/ 10640266.2019.1642035.

- Brewerton T. An overview of trauma-informed care and practice for eating disorders. J Aggress Maltreatment Trauma. 2018. https://doi.org/10. 1080/10926771.2018.1532940.
- Trottier K, Monson CM, Wonderlich SA, Olmsted MP. Initial Findings from project recover: overcoming co-occurring eating disorders and posttraumatic stress disorder through integrated treatment. J Trauma Stress. 2017;30(2):173–7. https://doi.org/10.1002/jts.22176.
- Akwa GGZ. Zorgstandaard Eetstoornissen; 2017. https://www.ggzst andaarden.nl/zorgstandaarden/eetstoornissen. Accessed 11 november 2017
- 14. Yager J, Devlin MJ, Halmi KA, Herzog DB, Mitchell JE, Powers P, Zerbe KJ. Guideline watch: Practice guideline for the treatment of patients with eating disorders (3e ed.); 2012.
- Abbate-Daga G, Buzzichelli S, Amianto F, Rocca G, Marzola E, McClintock SM, Fassino S. Cognitive flexibility in verbal and nonverbal domains and decision making in anorexia nervosa patients: a pilot study. BMC Psychiatry. 2011;11:1. https://doi.org/10.1186/ 1471-244X-11-162.
- Danner UN, Sanders N, Smeets PAM, Van Meer F, Adan RAH, Hoek HW, Van Elburg AA. Neuropsychological weaknesses in anorexia nervosa: set-shifting, central coherence, and decision making in currently ill and recovered women. Int J Eat Disord. 2012;45:5. https://doi.org/10.1002/ eat.22007.
- Martinez G, Cook-Darzens S, Chaste P, Mouren MC, Doyen C. L'anorexie mentale à la lumière du fonctionnement neurocognitif: nouvelles perspectives théoriques et thérapeutiques [Anorexia nervosa in the light of neurocognitive functioning: new theoretical and therapeutic perspectives]. L'Encephale. 2014;40:2. https://doi.org/10.1016/j.encep. 2012.06.004.
- Cavan J, Connan F. Eating disorders management. Clin Pharm. 2010;330–3.
- Brockmeyer T, Holtforth MG, Bents H, Kämmerer A, Herzog W, Friederich HC. Starvation and emotion regulation in anorexia nervosa. Compr Psychiatry. 2012;53:5. https://doi.org/10.1016/j.comppsych.2011.09.003.
- Harrison A, Sullivan S, Tchanturia K, Treasure J. Emotion recognition and regulation in anorexia nervosa. Clin Psychol Psychother. 2009;16:4. https://doi.org/10.1002/cpp.628.
- Ten Napel-Schutz MC, Vroling M, Mares SHW, Artnz A. Treating PTSD with Imagery Rescripting in underweight eating disorder patients: a multiple baseline case series study. J Eat Disord. 2022;10:35. https://doi. org/10.1186/s40337-022-00558-1.
- 22. Arntz A. Imagery Rescripting for PTSD related to childhood traumas: Treatment Protocol for the IREM study. Int Doc Univ Amst. 2015;1:1.
- Boterhoven de Haan KL, Lee CW, Fassbinder E, Van Es SM, Menninga S, Meewisse ML, Rijkeboer M, Arntz A. Imagery rescripting and eye movement desensitization and reprocessing as treatment for adults with post-traumatic stress disorder from childhood trauma: randomised clinical trial. Br J Psychiatry. 2020. https://doi.org/10.1192/bjp.2020.158.
- Morina N, Lancee J, Arntz A. Imagery rescripting as a clinical intervention for aversive memories: a meta-analysis. J Behav Therapy Exp Psychiatry. 2017. https://doi.org/10.1016/j.jbtep.2016.11.003.
- 25. Raabe S, Ehring T, Marquenie L, Olff M, Kindt M. Imagery Rescripting as stand-alone treatment for post-traumatic stress disorder related to childhood. J Behav Therapy Exp Psychiatry. 2015. https://doi.org/10.1016/j.jbtep.2015.03.013.
- Luber M. Eye movement desensitization and reprocessing (emdr) therapy scripted protocols and summary sheets. treating anxiety obsessive-compulsive and mood-related conditions; 2016. Springer Publishing Company LLC. Retrieved October 11 2022 from http://site. ebrary.com/id/11121132.
- Steil P, Lechner-Meichsner F, Johow J, Krüger-Gottschalk A, Mewes R, Reese JP, Schumm H, Weise C, Morina N, Ehring T. Brief imagery rescripting vs. usual care and treatment advice in refugees with posttraumatic stress disorder: study protocol for a multi-center randomized-controlled trial. Eur J Psychotraumatol. 2021;12(1):1872967. https://doi.org/10.1080/20008198.2021.1872967.
- Arntz A. Imagery rescripting as a therapeutic technique: review of clinical trials, basic studies, and research agenda. J Exp Psychopathol. 2012;
 3:2.doi.org/https://doi.org/10.5127/jep.024211.
- 29. Arntz A, Tiesema M, Kindt M. Treatment of PTSD: a comparison of imaginal exposure with and without imagery rescripting. J Behav Ther

- Exp Psychiatry. 2007;38(4):345–70. https://doi.org/10.1016/j.jbtep.2007. 10.006.
- 30. Green J, Thorogood N. Qualitative methods for health research. 4th ed. SAGE Publications; 2018. ISBN 9781473997103.
- 31. Guest G, Bunce A, Johnson L. How many interviews are enough? Field Methods. 2006;18:59–82.
- Abma TA, Nierse CJ, Widdershoven GAM. Patients as partners in responsive research: Methodological notions for collaborations in mixed research teams. Qual Health Res. 2009;19(3):401–15. https://doi. org/10.1177/1049732309331869.
- Nierse CJ, Schipper K, Van Zadelhoff E, Van de Griendt J, Abma TA. Collaboration and co-ownership in research: dynamics and dialogues between patient research partners and professional researchers in a research team. Health Expect. 2012;15(3):242–54. https://doi.org/10. 1111/j.1369-7625.2011.00661.
- Schipper K, Abma TA, Van Zadelhoff E, Van de Griendt J, Nierse CJ, Widdershoven GAM. What does it mean to be a patient research partner? Ethnodrama Qual Ing. 2010;16(6):501–10.
- 35. Frambach JM, Van der Vleuten CP, Durning SJ. AM last page: quality criteria in qualitative and quantitative research. Acad Med. 2013;88(4):552.
- Meadows LM, Morse JM. Constructing evidence within the qualitative project. In: Morse JM, Swanson JM, Kuzel AJ, editors. The nature of qualitative evidence. SAGE Publications. 2001. pp. 188–202.
- Olofsson ME, Oddli HW, Hoffart A, Eielsen HP, Vrabel KR. Change processes related to long-term outcomes in eating disorders with childhood trauma: an explorative qualitative study. J Couns Psychol. 2020;67(1):51–65. https://doi.org/10.1037/cou0000375.
- Eichhorn S, Brähler E, Franz M, Friedrich M, Glaesmer H. Traumatic experiences, alexithymia, and posttraumatic symptomatology: a crosssectional population-based study in Germany. Eur J Psychotraumatol. 2014;27:5. https://doi.org/10.3402/ejpt.v5.23870.
- Nowakowski ME, McFarlane T, Cassin S. Alexithymia and eating disorders: a critical review of the literature. J Eat Disord. 2013;1:21. https://doi.org/10.1186/2050-2974-1-21.
- Westwood H, Kerr-Gaffney J, Stahl D, Tchanturia K. Alexithymia in eating disorders: systematic review and meta-analyses of studies using the Toronto alexithymia scale. J Psychosom Res. 2017;99:66–81. https://doi.org/10.1016/j.jpsychores.2017.06.007.
- 41. Strauss C, Lever Taylor B, Gu J, Kuyken W, Baer R, Jones F, et al. What is compassion and how can we measure it? A review of definitions and measures. Clin Psychol Rev. 2016;1(47):15–27.
- Bosch M, Arntz A. Imagery rescripting for patients with posttraumatic stress disorder: a qualitative study of patients' and therapists' perspectives about the elements of change. Cogn Behav Pract. 2021. https:// doi.org/10.1016/j.cbpra.2021.08.001.
- Olofsson ME, Oddli HW, Vrabel KAR, Hoffart A. In solitude is safeness: a patient perspective on eating disorders in the context of multiple childhood trauma. Nordic Psychol. 2021;73(1):29–42. https://doi.org/ 10.1080/19012276.2020.1762714.
- Greenberg RP, Constantino MJ, Bruce N. Are patient expectations still relevant for psychotherapy process and outcome? Clin Psychol Rev. 2006;26(6):657–78. https://doi.org/10.1016/j.cpr.2005.03.002.
- Price M, Anderson P, Henrich CC, Rothbaum BO. Greater expectations: using hierarchical linear modeling to examine expectancy for treatment outcome as a predictor of treatment response. Behav Ther. 2008;39(4):398–405. https://doi.org/10.1016/j.beth.2007.12.002.
- Westra HA, Dozois DJA, Marcus M. Expectancy, homework compliance, and initial change in cognitive-behavioral therapy for anxiety. J Consult Clin Psychol. 2007;75(3):363–73. https://doi.org/10.1037/0022-006X. 75.3.363.
- Boterhoven de Haan KL, Lee CW, Correia H, Menninga S, Fassbinder E, Köehne S, et al. Patient and therapist perspectives on treatment for adults with PTSD from childhood trauma. J Clin Med. 2021;10(5):954. https://doi.org/10.3390/jcm10050954.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.